Author's response to reviews

Title: The efficacy of intraoperative methylene blue enemas to assess the integrity of a colonic anastomosis

Authors:

Stanton Smith MD (sts38@drexel.edu)
William McGeehin MD (wmcgeehin06759@yahoo.com)
Robert A. Kozol MD (kozol@nsq.uchc.edu)
David Giles MD (dgiles@nbgh.org)

Version: 5 Date: 30 July 2007

Author's response to reviews:

The following changes were implemented as requested along with various minor grammatical editing.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Qualifications - Please remove the authors’ qualifications (e.g. PhD / MD) from the manuscript and submissions system

Positions - Please remove the authors' positions (e.g. Surgery Resident, Surgeon, Chairman and Professor of Surgery, Assistant Professor of Clinical Surgery,) from the manuscript file.

Country - Please remember to include the country in the affiliation details.

Page 4 - Please delete the extra dot next to the text 'Windows. '

References - References should be cited in the text using consecutive numbers in square brackets. Please place the reference before the full stop. Please do not use superscript numbers.

We have noted that the references cited in the text started from 1, 2, 3 and then jumped to 8, 13, and then 10, 5, 11, etc. Please ensure the references cited in the text in the ascending numerical order.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.

If your references are formatted using EndNote or Refman, please make any changes in the EndNote or Refman record, rather than editing the references in the manuscript manually.

Figures -> Figure legends
Tables - Tables must be divided into cells/fields. Tables generated with tabbed text are not acceptable. Please ensure that texts are entered in separate rows and not separated by hard returns.

Table 1 legend - **including.... please use upper case, i.e. **Including....

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table.

Table 2 legends - Please left align the legends. Please run on the legends in one paragraph and separate them with a semi-colon.

Text in tables - Please may we recommend that either the abbreviations have their full names in the legends, or use the full names instead of abbreviations. Please use sentence case (i.e. remove all unnecessary capitals).